Author's response to reviews

**Title:** A systematic review with procedural assessments and meta-analysis of Low Level Laser Therapy in lateral elbow tendinopathy (tennis elbow)

**Authors:**

Jan M Bjordal (jmb@hib.no)  
Rodrigo AB Lopes-Martins (rmartins@icb.usp.br)  
Jon Joensen (jioe@hib.no)  
Anne E Ljunggren (elisabeth.ljunggren@isf.uib.no)  
Christian Couppe (ccou0001@bbhregionh.dk)  
Apostolos Stergioulas (asterg@uop.gr)  
Mark I Johnson (M.Johnson@leedsmet.ac.uk)

**Version:** 4 **Date:** 17 April 2008

**Author's response to reviews:** see over
Dear editor

This cover letter includes a point by point description of corrections and details on how the requested formatting changes have been incorporated into the manuscript.

1)

We have inserted the following Acknowledgements between the main body of the manuscript and the references:

Acknowledgements

The study was funded internally by Bergen University College in providing working hours for the first author, Jan M. Bjordal. The funding body played no role in the study design, the collection, analysis, interpretation of data, in the writing of the manuscript or the decision to submit the manuscript for publication. None of the other authors received any funding for the performance of the study.

2)

Permission has been received from Bergen University College to mention their name under acknowledgements.

3)

The manuscript has been revised and formatted according to the journal standards.

We hope everything now is in order and we are looking forward to see the paper published.

Best regards

Jan M. Bjordal